Author's response to reviews

Title: Clinical significance in the number of involved lymph nodes in patients that underwent surgery for pathological stage III-N2 non-small cell lung cancer

Authors:

Takeshi Hanagiri (hanagiri@med.uoeh-u.ac.jp)
Masaru Takenaka (masarururu_1230@yahoo.co.jp)
Sohich Oka (okasou@med.uoeh-u.ac.jp)
Yoshiki Shigematsu (yoshiki@med.uoeh-u.ac.jp)
Yoshika Nagata (y-nagata@med.uoeh-u.ac.jp)
Hidehiko Shimokawa (mikegonn@yahoo.co.jp)
Hidetaka Uramoto (hidetaka@med.uoeh-u.ac.jp)
Fumihiro Tanaka (tanaka@med.uoeh-u.ac.jp)

Version: 2 Date: 6 October 2011

Author's response to reviews:

Editors-in-Chief: Vipin Zamvar,
“Journal of Cardiothoracic Surgery”

October 5, 2011

Dear Professor Vipin Zamvar,
Re: MS: 5468292745831071

Title: Clinical significance in the number of involved lymph nodes in patients that underwent surgery for pathological stage III-N2 non-small cell lung cancer

We would be grateful if you could again review and consider our manuscript for publication in Journal of Cardiothoracic Surgery. The comments from the reviewers have been helpful in allowing us to revise our manuscript. We have addressed all of the comments and suggestions made by the reviewers. We have highlighted all changes in the revised manuscript to facilitate its review. We hope that our explanations and revisions are satisfactory, and that the revised version of our manuscript is now suitable for publication in Journal of Cardiothoracic Surgery. Thank you in advance for your kind consideration of our revised manuscript.

Sincerely yours,

Takeshi Hanagiri, MD.
Second Department of Surgery,
University of Occupational and Environmental Health,
Yahatanishi-ku, Kitakyushu, Fukuoka 807-8555, Japan.
Phone: +81(93) 603-1611
FAX: +81(93) 692-4004,
E-mail: hanagiri@med.uoeh-u.ac.jp